Math 153 Spring 2005 R. Schultz

UPDATED GENERAL INFORMATION — APRIL 7, 2005 Here is the second homework assignment, which is due in class on Wednesday, April 13, 2005. “Burton” refers to the course text by Burton.

Burton, p. 98: 3, 6, 11cd Burton, p. 112: 2, 3, 7, 18 Burton, p. 123: 4 Burton, p. 478: 3, 6, 11 Other information For the quiz on Tuesday, April 12, students should be prepared to convert ordinary fractions to Babylonian notation and vice versa in simple cases, and also to find Egyptian fraction expansions for fractions in standard form that are not especially complicated.